Author's response to reviews

Title: Quantifying behavioural interactions between humans and mosquitoes: Evaluating the protective efficacy of insecticidal nets against malaria transmission in rural Tanzania

Authors:

Gerry F Killeen (gkilleen@ihrdc.or.tz)
Japhet Kihonda (jkihonda@ihrdc.or.tz)
Edith Lyimo (lyimoe@ug.afro.who.int)
Fred R Oketch (fros2001@hotmail.com)
Maya E Kotas (maya.kotas@yale.edu)
Evan Mathenge (emathenge@mbita.mimcom.net)
Joanna A Schellenberg (joanna.schellenberg@lshtm.ac.uk)
Christian Lengeler (christian.lengeler@unibas.ch)
Thomas A Smith (thomas-a.smith@unibas.ch)
Chris J Drakeley (c.j.drakeley@lshtm.ac.uk)

Version: 3 Date: 1 November 2006

Author’s response to reviews:

We now submit this paper for publication by BMC Infectious Diseases and, in keeping with the comments of reviewer 1, have retained the sections on larviciding and ethical concerns regarding human landing catches. Minor typographical corrections have been included.